

Prevention of venous thromboembolic events in patients with lower leg immobilization after trauma: Systematic review and network meta-analysis with meta-epsidemiological approach

D. Douillet, C. Chapelle, E. Ollier, P. Mismetti, P.-M. Roy, Silvy Laporte

▶ To cite this version:

D. Douillet, C. Chapelle, E. Ollier, P. Mismetti, P.-M. Roy, et al.. Prevention of venous thromboembolic events in patients with lower leg immobilization after trauma: Systematic review and network meta-analysis with meta-epsidemiological approach. PLoS Medicine, 2022, 19 (7), pp.e1004059. 10.1371/journal.pmed.1004059 . hal-04805634

HAL Id: hal-04805634 https://hal.science/hal-04805634v1

Submitted on 16 Feb 2025

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



G OPEN ACCESS

Citation: Douillet D, Chapelle C, Ollier E, Mismetti P, Roy P-M, Laporte S (2022) Prevention of venous thromboembolic events in patients with lower leg immobilization after trauma: Systematic review and network meta-analysis with meta-epsidemiological approach. PLoS Med 19(7): e1004059. https://doi.org/10.1371/journal.pmed.1004059

Academic Editor: Donald A. Redelmeier, University of Toronto, CANADA

Received: February 7, 2022

Accepted: June 28, 2022

Published: July 18, 2022

Copyright: © 2022 Douillet et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All our data is available. The Documentary Search Strategies are in <u>S1 Table</u>. The statistical codes used are available on the packages mentioned in the article and the specific code used can be given on request. The data concerning the PRONOMOS study are available on specific request to the institutional data access committee (contact: innovte@chu-stetienne.fr). RESEARCH ARTICLE

Prevention of venous thromboembolic events in patients with lower leg immobilization after trauma: Systematic review and network metaanalysis with meta-epsidemiological approach

D. Douillet^{1,2,3}*, C. Chapelle⁴, E. Ollier^{4,5}, P. Mismetti^{3,4}, P.-M. Roy^{1,2,3}, S. Laporte^{3,4}

1 Emergency Department, Angers University Hospital, Health Faculty, Angers, France, 2 UNIV Angers, UMR MitoVasc CNRS 6015 –INSERM 1083, Equipe CARME, Angers, France, 3 F-CRIN INNOVTE network, Saint-Etienne, France, 4 Clinical Pharmacology Department, University Hospital of Saint-Etienne, Saint-Etienne, France, 5 SAINBIOSE INSERM U1059, Vascular Dysfunction and Hemostasis, Jean Monnet University, Saint-Etienne, France

* delphinedouillet@gmail.com

Abstract

Background

Lower limb trauma requiring immobilization is a significant contributor to overall venous thromboembolism (VTE) burden. The clinical effectiveness of thromboprophylaxis for this indication and the optimal agent strategy are still a matter of debate. Our main objective was to assess the efficacy of pharmacological thromboprophylaxis to prevent VTE in patients with isolated temporary lower limb immobilization after trauma. We aimed to estimate and compare the clinical efficacy and the safety of the different thromboprophylactic treatments to determine the best strategy.

Methods and findings

We conducted a systematic review and a Bayesian network meta-analysis (NMA) including all available randomized trials comparing a pharmacological thromboprophylactic treatment to placebo or to no treatment in patients with leg immobilization after trauma. We searched Medline, Embase, and Web of Science until July 2021. Only RCT or observational studies with analysis of confounding factors including adult patients requiring temporary immobilization for an isolated lower limb injury treated conservatively or surgically and assessing pharmacological thromboprophylactic agents or placebo or no treatment were eligible for inclusion. The primary endpoint was the incidence of major VTE (proximal deep vein thrombosis, symptomatic VTE, and pulmonary embolism-related death). We extracted data according to Preferred Reporting Items for Systematic Reviews and Meta-analyses for NMA and appraised selected trials with the Cochrane review handbook. Fourteen studies were included (8,198 patients). Compared to the control group, rivaroxaban, fondaparinux, and low molecular weight heparins were associated with a significant risk reduction of major VTE with an odds ratio of 0.02 (95% credible interval (CrI) 0.00 to 0.19), 0.22 (95% CrI 0.06 to 0.65), and 0.32 (95% CrI 0.15 to 0.56), respectively. No increase of the major bleeding

Funding: The authors received no specific funding for this work.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: -SL reports grants from Leo Pharma and personal fees from Eli Lilly and Bayer Healthcare, outside the submitted work. -PMR reports personal fees and non-financial support from Bayer Health care, from Boehringer Ingelheim France, from Bristol Myers Squibb, from Pfizer, from Aspen, from Daiichi Sankyo, and from Sanofi Aventis France, outside the submitted work.

Abbreviations: DOAC, direct oral anticoagulant; DVT, deep vein thrombosis; ISTH, International Society on Thrombosis and Haemostasis; LMWH, low molecular weight heparin; NMA, network metaanalysis; OR, odds ratio; PE, pulmonary embolism; RCT, randomized controlled trial; ROBINS-I, risk of bias in nonrandomized studies—of interventions; SUCRA, surface under the cumulative ranking curve; VTE, venous thromboembolism; 95% Crl, 95% credible interval. risk was observed with either treatment. Rivaroxaban has the highest likelihood of being ranked top in terms of efficacy and net clinical benefit. The main limitation is that the network had as many indirect comparisons as direct comparisons.

Conclusions

This NMA confirms the favorable benefit/risk ratio of thromboprophylaxis for patients with leg immobilization after trauma with the highest level of evidence for rivaroxaban.

Trial registration

PROSPERO CRD42021257669.

Author summary

Why was this study done?

- Lower limb trauma requiring immobilization is a significant contributor to overall venous thromboembolism (VTE) burden.
- Existing evidence suggests that pharmacological prophylaxis could reduce overall VTE event rates in these patients, but the proportional reduction of symptomatic events remains unclear as does the treatment that provides the best clinical benefit.
- International guidelines continue to offer conflicting advice for clinicians and thromboprophylaxis strategies vary widely between physicians in different countries.

What did the researchers do and find?

- This study is a systematic review and a Bayesian network meta-analysis including all available randomized trials comparing a pharmacological thromboprophylactic treatment to placebo or no treatment in patients with leg immobilization after trauma.
- We assess the efficacy and safety of 5 different existing strategies (LMWH, fondaparinux, aspirin, rivaroxaban, and control) to compare them against one another and to define the one that appears to be the most efficient with the highest level of evidence.
- Including 14 RCTs and 8,198 patients, rivaroxaban, fondaparinux, and low molecular weight heparins were associated with a significant risk reduction of major VTE.

What do these findings mean?

- This study confirms the efficacy of anticoagulant treatment to prevent VTE events in lower extremity trauma patients. It shows the potential value of rivaroxaban, this treatment having the best efficacy and safety profile.
- Additionally, it recommends that the efficacy of aspirin is uncertain and requires further well-performed investigation.

Introduction

Background

Patients with an isolated lower limb trauma requiring immobilization are at increased risk of venous thromboembolism (VTE), which includes deep vein thrombosis (DVT) (proximal or distal) and pulmonary embolism (PE). VTE is a serious and potentially life-threatening complication. In this population including various situations, asymptomatic VTE occurred in 18.0% (95% CI 12.9 to 23.1) and symptomatic VTE in 2.0% (95% CI 1.3 to 2.7) [1,2]. The efficacy and safety of thromboprophylaxis in reducing thromboembolic events in immobilized lower limb trauma patients has been established in several trials [3-6]. However, the POT--CAST study showed no significant effects of low molecular weight heparins (LMWHs) versus no treatment on the rate of symptomatic VTE [3]. Conversely, recent publications assessing direct oral anticoagulants (DOACs) and aspirin suggest that these molecules are effective [6,7]. Nowadays, guidelines on thromboprophylaxis and, therefore, clinical practices vary widely between countries, ranging from no preventive anticoagulation in the United States (unless the trauma is to be managed surgically) [8] to thromboprophylaxis for all patients for whom plantar support is not possible in France [9]. A number of countries, such as England, Germany, and Australia, propose that the risk of thrombosis should be assessed before initiating thromboprophylaxis [10-12]. It has been observed that there is no consensus among clinicians who do not follow the recommendations in a consensual manner in terms of deciding whether initiating thromboprophylaxis or in terms of choosing an antithrombotic molecule. In addition, the recommendations should be updated as new studies have been published [8].

Horner and colleagues published the only network meta-analysis (NMA) pooling the results of 13 randomized controlled trials (RCTs) on this population [13]. It concludes that LMWH and fondaparinux were effective in preventing any VTE in trauma patients requiring thromboprophylaxis. However, this review has some limitations. First, they assessed a 3-arm network with LMWH, fondaparinux and a control group pooling placebo, no treatment, and aspirin. But considering aspirin as an inactive molecule is questionable, given that aspirin is at the same time a recommended thromboprophylaxis for patients with major surgery mainly in the context of elective surgery (i.e., patients undergoing total hip or total knee arthroplasty) in the US [8,14,15]. Aspirin deserves to be singled out as a potential active treatment. Moreover, LMWH and fondaparinux were only compared with a control and not with each other. Second, the level of evidence varies greatly from study to study. Although they have shown no difference in treatment effect in a sensitivity analysis removing 3 studies considered at high risk of bias by the authors, the studies deserve to be analyzed separately according to the risk of bias. In addition, since this NMA was performed, new data are available with DOAC, especially with the publication of the PRONOMOS study, an RCT comparing rivaroxaban versus LMWH for patients with nonmajor surgery [6]. DOACs have proven their efficacy in major orthopedic surgery [16–18], and rivaroxaban, apixaban, and dabigatran are currently recommended for thromboprophylaxis after a total hip or knee replacement and should also be considered as a potential active treatment.

This systematic review coupled with an NMA aims to assess the efficacy of pharmacological thromboprophylaxis to prevent VTE in patients with temporary lower limb immobilization after trauma. Our main objective was to estimate the clinical efficacy and safety of each of the pharmacological thromboprophylaxis options and to compare the molecules to determine the best strategy available. We also assessed the potential modification of the treatment effect according to the risk of bias of the studies using a meta-epidemiological approach.

Materials and methods

Study protocol

This study is reported as per the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) guideline [19]. The protocol of the review was registered on the PROS-PERO international prospective register of systematic reviews (CRD42021257669).

Search strategy and selection criteria

An exhaustive literature search, both manual and computer assisted, was performed on all data up to and including July 29, 2021, with no restriction on the non-English languages in which the sources were published. The computer-assisted search was carried out on electronic databases (Medline, Embase), Google Scholar, the Cochrane Library, and the international database of clinical trials (www.clinicaltrials.gov). The following keywords were used: trauma, injury, immobilization, casts, Achilles' tendon rupture, LMWH, fondaparinux, DOAC, aspirin in combination with controlled or randomized, and trial, study, observational, and cohort study (S1 Table). Reference lists of journal articles, as well as proceedings from major international meetings (International Society on Thrombosis and Haemostasis, International Congress on Thrombosis), were manually reviewed to locate additional studies. Particular attention was paid to the risk of duplicate reports, and whenever identified, duplicate studies were excluded. When studies were published in both abstract form and as a full article, only the full publication was considered. When more than one article was available for the same study, the relevant information from all publications was extracted.

All references were reviewed for potential inclusion by 1 reviewer (DD), and any citations that clearly did not meet the inclusion criteria were excluded. All abstracts and full text articles were then examined independently by 2 reviewers (DD and CC). Any disagreements in the selection process were resolved through discussion or, if necessary, opinion of a third reviewer (SL). Disagreements were resolved by consensus, and all authors agreed on the final selection of included articles.

To be included in the meta-analysis, studies had to meet all the following criteria: (i) RCT or observational study with analysis of confounding factors; (ii) including adult patients requiring temporary immobilization (plaster cast or removable splint) for an isolated lower limb injury treated conservatively or surgically (excluding major polytrauma); (iii) comparing pharmacological thromboprophylactic agents with each other (i.e., LMWH agent, fondaparinux, aspirin, or DOAC) or with placebo or no treatment; and (iv) assessing the rate of VTE (DVT, PE, death related to PE) and/or major bleeding (as defined within each study). Studies published before 1990 were excluded. For studies including patients requiring temporary immobilization after trauma and/or for another reason as elective surgery, only the subgroup of patients with trauma were included in our meta-analysis. In this case, the authors were contacted to retrieve data from trauma patients.

Data extraction and quality assessment

Two of the authors (DD and CC) independently extracted data concerning study design, study quality, treatment regimens, population characteristics, and efficacy and safety outcomes. The data extracted from each study were reviewed and, in the case of disagreement, confirmed by arbitration by a third reviewer (SL). When data were unavailable or unclear, we attempted to contact the corresponding authors through e-mail and inspected previous systematic reviews for the trial data of interest. Any issues with data extraction were discussed and resolved by consensus.

Three authors (DD, CC, and SL) independently assessed the risk of bias within all included studies according to the Cochrane review handbook using on the Cochrane RoB 2 tool (Version 2 of the Cochrane risk-of-bias tool for randomized trials) taking into account random sequence generation, concealment of the allocation sequence, blinding of participants and personnel, blinding of outcome assessment, incomplete outcome data, and selective reporting, and ROBINS-I (risk of bias in nonrandomized studies—of interventions) for nonrandomized studies [20,21]. We considered studies as having a low risk of bias if the concealment of the allocation was adequate and the assessment of outcomes was blinded. In the case of older studies, there was a lack of protocol data and, therefore, imprecision in this item of the Rob2 tool, so it was not considered in the risk of bias assessment.

Outcome definitions

The primary efficacy endpoint was major VTE (DVT and/or PE) defined as the composite of symptomatic VTE, asymptomatic proximal DVT, and death related to PE assessed at the follow-up planned in each study. The secondary efficacy endpoint was all symptomatic VTE (DVT and/or PE) defined as reported within individual trials. In studies without systematic assessment of VTE (e.g., with compression leg vein ultrasonography or phlebography), VTE were classified as symptomatic VTE. The primary safety endpoint was major bleeding defined as reported in each study. Due to the inclusion of studies carried out before 2005, it was not possible to apply the criteria defined by the ISTH to define major bleeding [22]. For these studies, when bleeds were reported without specifying severity, they were considered as major bleeding. Finally, we assessed the net clinical benefit of each treatment defined as major VTE or major bleeding.

Statistical analyses

The control group included no treatment or placebo in the NMA. The different thromboprophylaxis drugs were considered as separate interventions (i.e., LMWH, aspirin, fondaparinux, and DOACs) in the NMA because of their different mechanisms of action. The different types of LMWH agent were pooled and considered as a single intervention. We made 2 network diagrams to illustrate which of the considered treatments (nodes) were compared (connected) directly and which were compared indirectly through one or more common comparators: the first including only RCT considered at low risk of bias and the second that including all RCTs. We conducted a Bayesian NMA with an unconstrained, random-effects model. Bayesian NMAs are commonly used as they naturally produce rankings and allow the use of prior distributions and direct probability statements can be made around Bayesian estimates, whereas frequentist methods rely on repeated sampling and *p*-values to inform conclusions. Events were supposed to follow a binomial distribution. We considered the possibility of a multi-arm study. Treatment effect estimates were presented as odds ratios (ORs) with 95% credible interval (95% CrI). Noninformative prior distributions were used for the model's parameters. The Gelman-Rubin statistics was used to test convergence [23]. The Gelman-Rubin statistics was checked automatically every set size number of iterations, and once the series had converged, we stored the last half of the sequence. The parameters tested for convergence were the relative treatment effects, baseline effect, and the heterogeneity parameter.

An inconsistency model was used to test the consistency assumption [24]. Inconsistency was evaluated using a node-splitting procedure. The inconsistency estimate and the corresponding *p*-value were reported. If the *p*-value was low, it meant that we could reject the null hypothesis that the direct and indirect evidence were consistent. We repeated this for all pairs in the network to identify pairs that might be inconsistent [25]. The node-splitting procedure

was applied for all loops in the network to identify ones that might be inconsistent. We also performed probabilistic analysis and report the results with a surface under the cumulative ranking curve (SUCRA), a numeric presentation of the overall ranking based on the probability that a treatment was most effective for the outcome of interest with the 95% CrI. Only thromboprophylactic agents that have demonstrated their efficacy have been included in this ranking.

We performed a subgroup analysis according to the risk of bias of the studies: low risk of bias, high risk of bias, or some concerns (i.e., with inadequate allocation concealment and/or nonblinded outcome assessment). We assessed the following potential treatment effect modifiers in a series of meta-regressions: age, duration of thromboprophylaxis, type of treatment (conservatively treated or surgical treatment), and the study design. To do so, we added continuous or discrete covariates to fit a network meta-regression. Statistical analyses were performed using R software version 4.0.3 (R Foundation for Statistical Computing) and the following R packages: "bnma," "rjags," and "ggplot2" [26,27].

Results

Selection of the studies

The study selection process is presented in Fig 1. The initial search identified 2,251 potentially eligible studies. Fourteen studies met the inclusion criteria. All were RCTs [3-6,28-37] including 8,198 patients. No observational studies were selected because none considered confounding bias in the analysis of the different treatments. Studies excluded after a full text review are listed in S2 Fig.

Description of included studies

Among the 14 RCTs included (8,198 patients), 7 studies were double blind (3,257 patients) [5,6,31,33-35,37], 4 were open label with a blind evaluation (PROBE [prospective randomized



Fig 1. PRISMA flow chart.

https://doi.org/10.1371/journal.pmed.1004059.g001

open trial with a blinded evaluation] (3,635 patients)) [3,4,32,36], and 3 were open without blind adjudication (1,306 patients) [28–30] (Table 1). All studies were published between 1993 to 2021. One German-language publication was translated [30]. Table 1 summarizes the key characteristics of the included studies. Thirteen studies assessed a LMWH (mainly dalteparin), 2 fondaparinux, 1 DOAC (rivaroxaban), and 1 aspirin. For 1 study with rivaroxaban involving patients with minor surgery, only the subgroup of patients with trauma were included in the present meta-analysis [6]. For the other studies, we considered the overall included populations. Seven studies focused on fracture or/and Achilles tendon rupture (n = 5,530 patients) [4,5,31,33–35,37]. Five studies included patients with conservative treatment (n = 2,850) [4,28,29,32,36], 6 with surgical treatments (n = 2,817) [5,6,33–35,37], and 3 with patients treated conservatively and/or surgically (n = 2,531) [3,30,31]. Finally, 7 studies included all lower limb trauma (i.e., fracture or soft tissue injury) (n = 2,668, 3) [6,28–30,32,36].

Quality assessment

The assessments of the risk of bias within each study are summarized in S1 and S2 Figs. Seven RCTs were considered at low risk of bias, i.e., with adequate allocation concealment and blinded outcome assessment (n = 3,962 patients) [3,6,29,33-35,37]. For the studies of Jorgensen and colleagues and Lassen and colleagues, information that would have assisted us in assessing bias was, however, missing. However, the authors were contacted during the meta-analysis of Zee and colleagues, enabling them to provide reliable data to assess the risk of bias in their article.

Network

Fig 2A and 2B display the network geometry of low risk of bias RCTs and all RCTs, respectively. The network including all RCTs was composed of a single closed loop consisting of 3 nodes, and 2 other direct comparisons. No studies have assessed other DOACs or warfarin in this indication.

Efficacy endpoints: Major VTE and symptomatic VTE

Thirteen trials reported outcomes for major VTE. Van Adrichem and colleagues did not perform a systematic assessment; as asymptomatic proximal DVT was not available, only symptomatic VTE were included for this study [3]. Without pharmacological thromboprophylaxis (control group), the rate of major VTE ranged from 0% to 11.7%, symptomatic VTE from 0% to 2.1%, and PE from 0% to 2.1%. Compared to the control group, rivaroxaban were associated with a significant risk reduction of major VTE in adults with lower limb immobilization after trauma (OR, 0.02; 95% CrI: 0.00 to 0.19), as well as fondaparinux (OR, 0.22; 95% CrI: 0.06 to 0.65) or LMWH (OR, 0.32, 95% CrI: 0.15 to 0.56) (Fig 3). No significant risk reduction of major VTE was shown with aspirin (OR, 0.13, 95% CrI: 0.00 to 2.22). When considering only low risk of bias studies, results are unchanged (Figs 4 and S3). No deaths were recorded in any of the studies except for the POT-CAST study, in which 1 death was reported to be related to PE.

The inconsistency of the model was tested on the direct comparison loop of LMWH, fondaparinux, and the control with a nonsignificant *p*-value (p = 0.335) allowing whole network estimates. Rivaroxaban is likely to be more effective than LMWH (OR, 0.07; 95% CrI: 0.00 to 0.56). This NMA did not detect a significant difference between the other treatments. Regarding the median rank, rivaroxaban is ranked first with a confidence interval between 1 to 2 (Fig 5). Presented in order, fondaparinux is ranked second (95% CrI 2 to 4) and LMWH is ranked third (95% CrI 2 to 4). When treatments were ranked, rivaroxaban had the highest likelihood

Author, Year Study acronym	Country	Design	Population	Treatment (conservative or surgical)	Intervention	Dose	Duration	Comparators	Assessment of VTE	No. of patients
Randomized con	trolled studies	5		8,	1					1
Kujath and colleagues, 1993 (25) 1993 (25)	Germany	Open	Patients over 16 years (mean age 34 years; female 42%), Trauma: fracture or soft tissue injury	Conservative treatment	LMWH- Nadroparin	2,850 IU	15 days	No treatment	US after plaster cast removal	306
Kock and colleagues, 1995 ^[26]	Germany	Open	Adults (18–75 years; mean age 34 years; female, 39%), Trauma: fracture or soft tissue injury	Conservative treatment	LMWH— certoparin	3,000 IU	14 days	No treatment	US confirmed by venography after plaster cast removal	428
Gehling and colleagues, 1998 —	Germany	Open	Patients over 16 years (mean age 36 years; female 51%), Trauma: fracture or soft tissue injury.	Management approach unclear (mainly surgically treated)	LMWH— Reviparin	1,750 IU	Throughout cast immobilization	Aspirin (1,000 mg/day)	Duplex sonography (all) or phlebography if thrombosis suspected)	572
Lassen and colleagues, 2002 ^[28]	Denmark	DB	Adults (>18 years; median age 47 years; female 48%), Trauma: fracture or Achilles tendon rupture.	Conservative or surgical treatment	LMWH— Reviparin	1,750 IU	43 days	Placebo	Unilateral venography after plaster cast removal	440
Jørgensen and colleagues, 2002 ^[29]	Denmark	PROBE	Adults (>18 years; mean age 48 years; female 43%), Trauma: fracture or soft-tissue injury.	Conservative treatment	LMWH— Tinzaparin	3,500 IU	38 days	No treatment	Unilateral venography after plaster cast removal	300
Lapidus and colleagues, 2007a ^[30]	Sweden	DB	Adults (18–75 years; mean age 40 years; female 21%), Trauma: Achilles tendon rupture.	Surgical treatment	LMWH— Dalteparin	5,000 IU	43 days	Placebo	Unilateral US confirmed by venography 3 and 6 weeks after surgery	105
Lapidus and colleagues, 2007b ^[31]	Sweden	DB	Adults (18–75 years; mean age 48 years; female 54%), Trauma: ankle fracture.	Surgical treatment	LMWH— Dalteparin	5,000 IU	44 days	Placebo	Unilateral venography at the end of treatment	272

Table 1. Characteristics of included studies and patients' characteristics.

(Continued)

Author, Year Study acronym	Country	Design	Population	Treatment (conservative or surgical)	Intervention	Dose	Duration	Comparators	Assessment of VTE	No. of patients
Goel and colleagues, 2009 ^[32]	Canada	DB	Adults (18–75 years; mean age 41 years; female 38%), Trauma: fracture.	Surgical treatment	LMWH— Dalteparin	5,000 IU	14 days	Placebo	Bilateral venography at the end of treatment	305
Samama and colleagues, 2013 ^[33] FONDACAST study	France	PROBE	Adults (>18 years; mean age 46 years; female 53.4%), Trauma: fracture or soft tissue injury.0/0/00 0:00:00 AM	Conservative treatment	Fondaparinux	2.5 mg	33.7 days	LMWH– Nadroparin (2,850 IU/day)	Compression ultrasonography and/or venography performed for suspected DVT after cast removal	1,349
Selby et al., 2015 [^{34]} <i>D-KAF study</i>	Canada	DB	Patients over 16 years (mean age 49 years; female 48%), Trauma: fracture.	Surgical treatment	LMWH— Dalteparin	5,000 IU	14 days	Placebo	Bilateral proximal US at end of treatment	265
Zheng and colleagues, 2016 ^[5]	China	DB	Adults (>18 years; mean age 47.8 years; female 37.7%), Trauma: fracture.	Surgical treatment	LMWH-no information	unknown	14 days	Placebo	Blinded bilateral Doppler compression ultrasound	814
Bruntink and colleagues, 2017 ^[4]	the Netherlands	PROBE	Adults (>18 years; mean age 47 years; female 58%), Trauma: fracture of the ankle or foot.	Conservative treatment	LMWH— Nadroparin	2,850 IU	40 days	Fondaparinux (2.5 mg/day) or no treatment	Duplex sonography after the removal of the cast	467
Van Adrichem and colleagues, 2017 ^[3] POT-CAST study	the Netherlands	Open	Adults (>18 years; mean age 46 years; female 50.1%), Trauma: fracture or soft tissue injury.	Conservative or surgical treatment	LMWH– Nadroparin or Dalteparin	2,850 IU	26 days	No treatment	Symptomatic VTE within 3 months after the procedure. DVT determined by abnormal compression US	1519
Samama and colleagues, 2020 ^[6] <i>PRONOMOS</i> <i>study</i>	France	DB	Subgroup analysis: only patients with trauma where selected. Adults (>18 years; mean age 46.1 years; female 36.7%),	Surgical treatment	DOAC— Rivaroxaban	10 mg	37 days	Enoxaparin (4,000 IU/day)	Compression ultrasonography at the end of the immobilization	1,056

Table 1. (Continued)

DB, double blind; DOAC, direct oral anticoagulant; DVT, deep vein thrombosis; IU, international unit; LMWH, low molecular weight heparin; PROBE, Prospective Randomized Open-label Blinded End-point; US, ultrasound; VTE, venous thromboembolism.

https://doi.org/10.1371/journal.pmed.1004059.t001





of being ranked top in terms of efficacy (SUCRA = 94.9%), followed by fondaparinux (SUCRA = 52.6%) and LMWH (SUCRA = 37.7%). Aspirin was not considered effective and was not included in this ranking. The results of the network meta-regressions highlighted that none of the covariates (age, mean duration of treatment, rate of patients surgically treated, and study design: double blind versus others) improved the fit of the model in these analyses, which suggests that they did not modify the treatment effects.

In the sensitivity analysis with the symptomatic VTE endpoint, the magnitude effect was similar for the major VTE endpoint (i.e., for rivaroxaban (OR, 0.05; 95% CrI: 0.00 to 0.49), fondaparinux (OR, 0.19; 95% CrI: 0.03 to 0.97), and LMWH (OR, 0.41, 95% CrI: 0.12 to 0.98) (S4 Fig).

Safety endpoint (major bleeding)

Data on major bleeding were available for 13 studies (n = 7,892). A total of 6 events occurred: none in the rivaroxaban, aspirin, and control groups, 5 among 3,556 patients in the LMWH group, and 1 among 3,466 patients in the fondaparinux group. A total of 9 patients had a nonmajor clinically relevant bleeding: 3 among 1,078 patients in the rivaroxaban group, 5 among 3,353 patients among the LMWH group, and 1 among 766 patients in the fondaparinux group. Given the very low risk of major bleeding, no model was built to evaluate this endpoint and no comparison of treatment by ranking was performed.

Net clinical benefit (major VTE and major bleeding)

Compared to the control group, rivaroxaban was associated with an increase of net clinical benefit in adult with lower limb immobilization after trauma (OR, 0.02; 95% CrI: 0.00 to 0.15), as well as fondaparinux (OR, 0.25; 95% CrI: 0.07 to 0.74) and LMWH (OR, 0.34, 95% CrI: 0.17



Fig 3. Network forest plot for the primary outcome (major VTE) in all RCTs with OR (points) and their 95% CrIs (lines). ASA, aspirin; DOAC, direct oral anticoagulant; LMWH, low molecular weight heparin; OR, odds ratio; RCT, randomized controlled trial; VTE, venous thromboembolism; 95% CrI, 95% credible interval.

to 0.63) (Fig 6). Rivaroxaban was likely to be more effective than LMWH (OR, 0.05, 95% CrI: 0.00 to 0.41) and fondaparinux (OR, 0.07, 95% CrI: 0.00 to 0.72).

Discussion

This NMA including 14 studies evaluating the efficacy and the safety of 4 anticoagulant treatments in the prevention of venous thromboembolic events in more than 8,000 patients immobilized after trauma shows that rivaroxaban, fondaparinux, and LMWHs reduce the risk of major VTE compared with a placebo or no treatment. Considering only symptomatic VTE, the results remain unchanged but with wider confidence intervals due to reduced power. These results were unchanged when including only low risk of bias studies, even if the magnitude of the effect of LMWH is less convincing. These results about LMWH are consistent with the other meta-analyses [38,39].

Of the DOACs, only rivaroxaban has been evaluated. This treatment was ranked first in terms of efficacy with no statistical excess risk of bleeding. A Bayesian NMA evaluating different thromboprophylaxis during hip or knee replacement surgery showed an identical ranking of molecules with rivaroxaban in the lead [40].

Prophylaxis in patients with lower extremity trauma is a particular and rather unique situation. The net clinical benefit endpoint is dominated by the risk of major VTE, the rate of major bleeding being much lower than the risk of VTE. That is not very surprising since the

OR (95% Crl)



Fig 4. Network forest plot according to the risk of bias for the main outcome: major VTE with OR (points) and their 95% CrIs (lines). DOAC, direct oral anticoagulant; LMWH, low molecular weight heparin; OR, odds ratio; RCT, randomized controlled trial; VTE, venous thromboembolism; 95% CrI, 95% credible interval.

https://doi.org/10.1371/journal.pmed.1004059.g004

median age is around 48 years, whereas in other types of primary prevention of VTE, patients are and therefore more at risk of bleeding but also because their clinical conditions are not associated with bleeding risk such as major surgical procedures, hospitalization for acute disease, or intensive care units.

The benefit-risk profile of these treatments is to date not questionable, except in terms of cost-effectiveness and patient preference. And beyond the question of the benefit that seems to be in favor, it is now necessary to define which one to choose.

The place of aspirin for thromboprophylaxis of patients with immobilization after injury remains a matter of debate. It is therefore not possible to answer its place in the therapeutic arsenal. Aspirin is indicated for primary prophylaxis in patients undergoing major orthopedic surgery (i.e., total hip or knee replacement) [15,41]. Aspirin must now be evaluated in this indication of patients with immobilization in the context of trauma. Regarding rivaroxaban, PRONOMOS was the first study to assess their efficacy and safety in nonmajor surgery patients. Apart from 2 observational studies with methodological weaknesses, there is no study evaluating DOAC in this large population of lower extremity trauma patients requiring immobilization without surgery [7,42]. In sum, although the ranking allows to compare the effect of the 3 different agents that were significantly effective versus control (rivaroxaban, LMWH,





and fondaparinux), this ranking remains imprecise. High-level evidence studies, mainly for oral treatments, are now necessary to guide future guidelines in this population [43,44].

This NMA is an update of the first one performed by Horner and colleagues. The creation of a living meta-analysis on this topic would be very useful to carry on these analyses. Living network meta-analyses allow for a broad, comprehensive and up-to-date presentation of evidence [45,46]. This living meta-analysis could lead to living clinical guidelines [46].

The main strength of this study is that it has synthesized data from more than 8,000 participants in 14 RCTs. This represents a large, methodologically robust method to simultaneously estimate of relative treatment effects. Furthermore, the analysis was performed with a Bayesian approach allowing the comparison and ranking of treatments between them. The analysis of individual data from the PRONOMOS study provided the first opportunity to assess DOAC in this indication through a high-level evidence RCT. The meta-analysis provides a global answer, but in the era of personalized treatment, patient preference and physician experience remain important in deciding whether or not to use prophylaxis based on baseline risk, estimated by medical experience or a risk score [1,44,47], and depending on the type of surgery and patient characteristics.

Some limitations need to be discussed. First, among the 14 studies performed in patients with lower extremity immobilization following trauma and eligible for this NMA, only 50% of the studies are considered at low risk of bias (i.e., with adequate allocation concealment and blinded outcome assessment). However, we reported small differences in terms of treatment effects estimated from RCTs at low risk of bias and all RCTs. Another limitation is that the network had as many indirect comparisons as direct comparisons. For example, aspirin is only linked by an indirect comparison to DOAC, fondaparinux, and a control. To date, there are





few data on the efficacy of aspirin in this indication (144 patients). Furthermore, the definitions of major VTE were not consensual and the choice of this definition may have underestimated the event rate in both arms of the studies that considered only symptomatic events (i.e., did not assess asymptomatic proximal DVT). Finally, among DOAC, only rivaroxaban was assessed, and our findings may be extrapolated to others, albeit with caution.

Conclusions

This NMA shows the efficacy of rivaroxaban, fondaparinux, and LMWH in preventing VTE in patients with lower extremity trauma requiring immobilization, compared with placebo or no treatment. Rivaroxaban has the highest likelihood of being top ranked in terms of efficacy and net clinical benefit. However, the decision to initiate thromboprophylaxis must also consider cost-effectiveness, cost-utility, baseline risk of each patient, estimated by medical experience or a risk score and patient preference.

Supporting information

S1 PRISMA Checklist. Preferred Reporting Items for Systematic reviews and Meta-Analyses extension for Scoping Reviews (PRISMA-ScR) Checklist. (DOCX) S1 Fig. Risk of bias assessment summary. (PDF)

S2 Fig. Risk of bias assessment graph. (PDF)

S3 Fig. Median rank and SUCRA values of competing prophylactic treatments for only RCT with low risk of bias for the primary outcome (major VTE). RCT, randomized controlled trial; SUCRA, surface under the cumulative ranking curve; VTE, venous thromboembolism. (PDF)

S4 Fig. Network Forest plot for the secondary outcome (symptomatic VTE) in all RCTs with OR (points) and their 95% CrIs (lines). ASA, aspirin; DOAC, direct oral anticoagulant; LMWH, low molecular weight heparin; OR, odds ratio; RCT, randomized controlled trial; VTE, venous thromboembolism; 95% CrI, 95% credible interval. (PDF)

S1 Table. Literature search strategies. (DOCX)

S2 Table. Excluded studies with rationale. (DOCX)

Acknowledgments

The authors thank Prof. Marc Samama for agreeing to the use of the individual data from the PRONOMOS study.

Author Contributions

Conceptualization: D. Douillet, C. Chapelle, S. Laporte.

Data curation: D. Douillet, S. Laporte.

Formal analysis: D. Douillet, C. Chapelle.

Investigation: D. Douillet.

Methodology: D. Douillet, C. Chapelle, P. Mismetti, P.-M. Roy, S. Laporte.

Project administration: D. Douillet, C. Chapelle, E. Ollier, S. Laporte.

Resources: D. Douillet.

Supervision: P. Mismetti, P.-M. Roy, S. Laporte.

Validation: C. Chapelle, E. Ollier, P.-M. Roy, S. Laporte.

Visualization: D. Douillet, P. Mismetti, P.-M. Roy, S. Laporte.

Writing - original draft: D. Douillet.

Writing – review & editing: D. Douillet, C. Chapelle, E. Ollier, P. Mismetti, P.-M. Roy, S. Laporte.

References

 Nemeth B, Cannegieter SC. Venous thrombosis following lower-leg cast immobilization and knee arthroscopy: From a population-based approach to individualized therapy. Thromb Res. 2019; 174:62– 75. https://doi.org/10.1016/j.thromres.2018.11.030 PMID: 30579148

- Grewal K, Atzema CL, Sutradhar R, Everett K, Horner D, Thompson C, et al. Venous Thromboenbolism in Patients Discharged From the Emergency Department With Ankle Fractures: A Population-Based Cohort Study. Ann Emerg Med. 2021;S0196-0644(21):00517–5. https://doi.org/10.1016/j. annemergmed.2021.06.017 PMID: 34535301
- van Adrichem RA, Nemeth B, Algra A, le Cessie S, Rosendaal FR, Schipper IB, et al. Thromboprophylaxis after Knee Arthroscopy and Lower-Leg Casting. N Engl J Med. 2017; 376:515–25. https://doi.org/ 10.1056/NEJMoa1613303 PMID: 27959702
- Bruntink MM, Groutars YME, Schipper IB, Breederveld RS, Tuinebreijer WE, Derksen RJ, et al. Nadroparin or fondaparinux versus no thromboprophylaxis in patients immobilised in a below-knee plaster cast (PROTECT): A randomised controlled trial. Injury. 2017; 48:936–40. https://doi.org/10.1016/j. injury.2017.02.018 PMID: 28279428
- Zheng X, Li D-Y, Wangyang Y, Zhang X-C, Guo K-J, Zhao F-C, et al. Effect of Chemical Thromboprophylaxis on the Rate of Venous Thromboembolism After Treatment of Foot and Ankle Fractures. Foot Ankle Int. 2016; 37:1218–24. https://doi.org/10.1177/1071100716658953 PMID: 27521353
- Samama CM, Laporte S, Rosencher N, Girard P, Llau J, Mouret P, et al. Rivaroxaban or Enoxaparin in Nonmajor Orthopedic Surgery. N Engl J Med. 2020; 382:1916–25. <u>https://doi.org/10.1056/ NEJMoa1913808 PMID: 32223113</u>
- Haque S, Davies MB. Oral thromboprophylaxis in patients with ankle fractures immobilized in a below the knee cast. Foot Ankle Surg. 2015; 21:266–8. https://doi.org/10.1016/j.fas.2015.02.002 PMID: 26564729
- Falck-Ytter Y, Francis CW, Johanson NA, Curley C, Dahl OE, Schulman S, et al. Prevention of VTE in Orthopedic Surgery Patients. Chest. 2012; 141:e278S–325S. https://doi.org/10.1378/chest.11-2404 PMID: 22315265
- Samama C-M, Gafsou B, Jeandel T, Laporte S, Steib A, Marret E, et al. French Society of Anaesthesia and Intensive Care. Guidelines on perioperative venous thromboembolism prophylaxis. Update 2011. Short text. Ann Fr Anesth Reanim. 2011; 30:947–51. https://doi.org/10.1016/j.annfar.2011.10.008 PMID: 22104443
- Roberts C, Horner D, Coleman G, Maitland L, Curl-Roper T, Smith R, et al. Guidelines in Emergency Medicine Network (GEMNet): guideline for the use of thromboprophylaxis in ambulatory trauma patients requiring temporary limb immobilisation. Emerg Med J. 2013; 30:968–82. https://doi.org/10.1136/ emermed-2013-203134 PMID: 24142947
- Encke A, Haas S, Kopp I. The Prophylaxis of Venous Thromboembolism. Dtsch Arztebl Int. 2016; 113:532–8. https://doi.org/10.3238/arztebl.2016.0532 PMID: 27581506
- Tran HA, Gibbs H, Merriman E, Curnow JL, Young L, Bennett A, et al. New guidelines from the Thrombosis and Haemostasis Society of Australia and New Zealand for the diagnosis and management of venous thromboembolism. Med J Aust. 2019; 210:227–35. https://doi.org/10.5694/mja2.50004 PMID: 30739331
- Horner D, Stevens JW, Pandor A, Nokes T, Keenan J, de Wit K, et al. Pharmacological thromboprophylaxis to prevent venous thromboembolism in patients with temporary lower limb immobilization after injury: systematic review and network meta-analysis. J Thromb Haemost. 2019. <u>https://doi.org/10. 1111/jth.14666</u> PMID: 31654551
- Hood BR, Cowen ME, Zheng HT, Hughes RE, Singal B, Hallstrom BR. Association of Aspirin With Prevention of Venous Thromboembolism in Patients After Total Knee Arthroplasty Compared With Other Anticoagulants: A Noninferiority Analysis. JAMA Surg. 2019; 154:65–72. https://doi.org/10.1001/ jamasurg.2018.3858 PMID: 30347089
- Anderson DR, Morgano GP, Bennett C, Dentali F, Francis CW, Garcia DA, et al. American Society of Hematology 2019 guidelines for management of venous thromboembolism: prevention of venous thromboembolism in surgical hospitalized patients. Blood Adv. 2019; 3:3898–944. <u>https://doi.org/10. 1182/bloodadvances.2019000975</u> PMID: 31794602
- Eriksson BI, Borris LC, Friedman RJ, Haas S, Huisman MV, Kakkar AK, et al. Rivaroxaban versus Enoxaparin for Thromboprophylaxis after Hip Arthroplasty. N Engl J Med. 2008; 358:2765–75. <u>https:// doi.org/10.1056/NEJMoa0800374 PMID: 18579811</u>
- Eriksson BI, Dahl OE, Rosencher N, Clemens A, Hantel S, Feuring M, et al. Oral dabigatran etexilate versus enoxaparin for venous thromboembolism prevention after total hip arthroplasty: pooled analysis of two phase 3 randomized trials. Thrombosis J. 2015; 13:36. https://doi.org/10.1186/s12959-015-0067-8 PMID: 26578849
- Neumann I, Rada G, Claro JC, Carrasco-Labra A, Thorlund K, Akl EA, et al. Oral direct Factor Xa inhibitors versus low-molecular-weight heparin to prevent venous thromboembolism in patients undergoing total hip or knee replacement: a systematic review and meta-analysis. Ann Intern Med. 2012; 156:710–9. https://doi.org/10.7326/0003-4819-156-10-201205150-00421 PMID: 22412038

- Page MJ, McKenzie JE, Bossuyt PM, Boutron I, Hoffmann TC, Mulrow CD, et al. statement: an updated guideline for reporting systematic reviews. BMJ. 2020; 2021:n71.
- Sterne JAC, Savović J, Page MJ, Elbers RG, Blencowe NS, Boutron I, et al. RoB 2: a revised tool for assessing risk of bias in randomised trials. BMJ. 2019; 366:I4898. <u>https://doi.org/10.1136/bmj.I4898</u> PMID: 31462531
- Sterne JA, Hernán MA, Reeves BC, Savović J, Berkman ND, Viswanathan M, et al. ROBINS-I: a tool for assessing risk of bias in non-randomised studies of interventions. BMJ. 2016; 355:i4919. <u>https://doi.org/10.1136/bmj.i4919</u> PMID: 27733354
- 22. Schulman S, Kearon C, Subcommittee on control of anticoagulation of the scientific and standardization committee of the international society on thrombosis and haemostasis. Definition of major bleeding in clinical investigations of antihemostatic medicinal products in non-surgical patients: Definitions of major bleeding in clinical studies. J Thromb Haemost. 2005; 3:692–4. https://doi.org/10.1111/j.1538-7836. 2005.01204.x PMID: 15842354
- Gelman A, Rubin DB. Inference from iterative simulation using multiple sequences. Stat Sci. 1992; 7:457–72.
- Higgins JPT, Jackson D, Barrett JK, Lu G, Ades AE, White IR. Consistency and inconsistency in network meta-analysis: concepts and models for multi-arm studies. Res Synth Methods. 2012; 3:98–110. https://doi.org/10.1002/jrsm.1044 PMID: 26062084
- Dias S, Welton NJ, Caldwell DM, Ades AE. Checking consistency in mixed treatment comparison metaanalysis. Stat Med. 2010; 29:932–44. https://doi.org/10.1002/sim.3767 PMID: 20213715
- Seo M, Schmid C. bnma: Bayesian Network Meta-Analysis using "JAGS." 2022. Available from: https:// cran.r-project.org/web/packages/bnma/bnma.pdf.
- Plummer M, Stukalov A, Denwood M. Package 'rjags': Bayesian Graphical Models using MCMC. Available from: https://cran.r-project.org/web/packages/rjags/rjags.pdf
- Kujath P, Spannagel U, Habscheid W. Incidence and prophylaxis of deep venous thrombosis in outpatients with injury of the lower limb. Haemostasis. 1993; 23(Suppl 1):20–6. <u>https://doi.org/10.1159/</u> 000216905 PMID: 8388353
- Kock H-J, Schmit-Neuerburg KP, Hanke J, Rudofsky G. Thromboprophylaxis with low-molecular-weight heparin in outpatients with plaster-cast immobilisation of the leg. Lancet. 1995; 346:459–61. <u>https://doi.org/10.1016/s0140-6736(95)91320-3 PMID: 7637478</u>
- Gehling H, Giannadakis K, Lefering R, Hessmann M, Achenbach S, Gotzen L. [Prospective randomized pilot study of ambulatory prevention of thromboembolism. 2 times 500 mg aspirin (ASS) vs. clivarin 1750 (NMH)]. Unfallchirurg. 1998; 101:42–9. https://doi.org/10.1007/s001130050231 PMID: 9522671
- Lassen MR, Borris LC, Nakov RL. Use of the low-molecular-weight heparin reviparin to prevent deepvein thrombosis after leg injury requiring immobilization. N Engl J Med. 2002; 347:726–30. <u>https://doi.org/10.1056/NEJMoa011327 PMID: 12213943</u>
- Jørgensen PS, Warming T, Hansen K, Paltved C, Vibeke Berg H, Jensen R, et al. Low molecular weight heparin (Innohep) as thromboprophylaxis in outpatients with a plaster cast: a venografic controlled study. Thromb Res. 2002; 105:477–80. https://doi.org/10.1016/s0049-3848(02)00059-2 PMID: 12091045
- Lapidus LJ, Ponzer S, Elvin A, Levander C, Lärfars G, Rosfors S, et al. Prolonged thromboprophylaxis with Dalteparin during immobilization after ankle fracture surgery: a randomized placebo-controlled, double-blind study. Acta Orthop. 2007; 78:528–35. https://doi.org/10.1080/17453670710014185 PMID: 17966008
- Lapidus LJ, Rosfors S, Ponzer S, Levander C, Elvin A, Larfars G. Prolonged Thromboprophylaxis With Dalteparin After Surgical Treatment of Achilles Tendon Rupture: A Randomized. Placebo-Controlled Study J Orthop Trauma. 2007; 21:6. https://doi.org/10.1097/01.bot.0000250741.65003.14 PMID: 17211270
- Goel DP, Buckley R, deVries G, Abelseth G, Ni A, Gray R. Prophylaxis of deep-vein thrombosis in fractures below the knee: a prospective randomised controlled trial. J Bone Joint Surg Br. 2009;91-B:388– 94. https://doi.org/10.1302/0301-620X.91B3.20820 PMID: 19258618
- 36. Samama C, Lecoules N, Kierzek G, Claessens Y, Riou B, Rosencher N, et al. Comparison of fondaparinux with low-molecular-weight heparin for VTE prevention in patients requiring rigid or semi-rigid immobilization for isolated non-surgical below-knee injury. J Thromb Haemost. 2013; 11(10):1833–43. https://doi.org/10.1111/jth.12395 PMID: 23965181
- Selby R, Geerts WH, Kreder HJ, Crowther MA, Kaus L, Sealey F. A Double-Blind, Randomized Controlled Trial of the Prevention of Clinically Important Venous Thromboembolism After Isolated Lower Leg Fractures. J Orthop Trauma. 2015; 29:224–30. https://doi.org/10.1097/BOT.0000000000250 PMID: 25900749

- Hickey BA, Watson U, Cleves A, Alikhan R, Pugh N, Nokes L, et al. Does thromboprophylaxis reduce symptomatic venous thromboembolism in patients with below knee cast treatment for foot and ankle trauma? A systematic review and meta-analysis. Foot Ankle Surg. 2018; 24:19–27. https://doi.org/10. 1016/j.fas.2016.06.005 PMID: 29413769
- **39.** Zee AA, van Lieshout K, van der Heide M, Janssen L, Janzing HM. Low molecular weight heparin for prevention of venous thromboembolism in patients with lower-limb immobilization. The Cochrane Library, John Wiley & Sons, Ltd; 2017.
- 40. Lewis S, Glen J, Dawoud D, Dias S, Cobb J, Griffin XL, et al. Venous thromboembolism prophylaxis strategies for people undergoing elective total knee replacement: a systematic review and network meta-analysis. Lancet Haematol. 2019; 6:e530–9. https://doi.org/10.1016/S2352-3026(19)30155-3 PMID: 31444124
- Afshari A, Ageno W, Ahmed A, Duranteau J, Faraoni D, Kozek-Langenecker S, et al. European Guidelines on perioperative venous thromboembolism prophylaxis: Executive summary. Eur J Anaesthesiol. 2018; 35:77–83. https://doi.org/10.1097/EJA.00000000000729 PMID: 29112553
- Ali IA, Elbana H, Farhan M, Qureshi IF, Mubashir A. Venous thromboembolism in trauma patients with lower limb cast immobilization, associated risk reduction and complication using rivaroxaban. Ir J Med Sci. 2021; 190:169–75. https://doi.org/10.1007/s11845-020-02306-3 PMID: 32642982
- Douillet D, Penaloza A, Horner D, Savary D, Hugli O, Nemeth B, et al. Evidence-based guidelines for thromboprophylaxis in patients with lower limb trauma requiring immobilization: an urgent, unmet need. Eur J Emerg Med. 2020; 27:245–6. https://doi.org/10.1097/MEJ.0000000000000677 PMID: 32618770
- Nemeth B, Douillet D, le Cessie S, Penaloza A, Moumneh T, Roy P-M, et al. Clinical risk assessment model to predict venous thromboembolism risk after immobilization for lower-limb trauma. EClinicalMedicine. 2020; 20:100270. https://doi.org/10.1016/j.eclinm.2020.100270 PMID: 32300734
- Elliott JH, Synnot A, Turner T, Simmonds M, Akl EA, McDonald S, et al. Living systematic review: 1. Introduction-the why, what, when, and how. J Clin Epidemiol. 2017; 91:23–30. <u>https://doi.org/10.1016/j.jclinepi.2017.08.010</u> PMID: 28912002
- Bartoszko JJ, Siemieniuk RAC, Kum E, Qasim A, Zeraatkar D, Ge L, et al. Prophylaxis against covid-19: living systematic review and network meta-analysis. BMJ. 2021; 373:n949. https://doi.org/10.1136/ bmj.n949 PMID: 33903131
- Douillet D, Nemeth B, Penaloza A, Le Gal G, Moumneh T, Cannegieter SC, et al. Venous thromboembolism risk stratification for patients with lower limb trauma and cast or brace immobilization. PLoS ONE. 2019; 14:e0217748. https://doi.org/10.1371/journal.pone.0217748 PMID: 31220097